Author’s response to reviews

Title: Community Mobilization and Maternal Care of Women Living With HIV in poor settings: The case of Mfuwe, Zambia

Authors:

Choolwe Muzyamba (muzyamba@merit.unu.edu)
Wim Groot (w.groot@maastrichtuniversity.nl)
Sonila Tomini (tomini@merit.unu.edu)
Milena Pavlova (m.pavlova@maastrichtuniversity.nl)

Version: 4 Date: 21 Feb 2018

Author’s response to reviews:

Dear Editor,

RE BHSR-D-16-00966R2

Community Mobilization and Maternal Care of Women Living With HIV in poor settings: The case of Mfuwe, Zambia

Thank you for correcting and providing comments on our paper. We have accepted all your suggestions and adjusted our paper accordingly. Below you will find our specific responses to your comments.

1. As your manuscript does not contain any individual person’s identifiable data or information, please remove your current statement and state “Not applicable” in the ‘Consent to Publish’ section.

-We have adjusted this as such
2. Please rename your ‘Methodology’ heading in the Abstract to ‘Methods’. Please also ensure that you make these changes to both Abstracts (the one within the Manuscript and the one within the Editorial Submission System) as these need to match.

-We have renamed it in both abstracts

3. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

-We have provided the list and provided explanation at first use in the main text

4. Unfortunately, we still believe that Table 2 contains far too many indirect identifiers, which could potentially compromise the anonymity of the participants. Please remove the ‘Marital status’ column. Apologies for the inconvenience.

-We have adjusted as such

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

-We have provided it as instructed